Author's response to reviews

Title: Teaching Evidence-Based Complementary and Alternative Medicine (EBCAM); Changing behaviours in the face of reticence: A cross-over trial

Authors:
Edward J Mills (emills@ccnm.edu)
Taras T Hollyer (thollyer@ccnm.edu)
Ron Saranchuk (rsaranchuk@ccnm.edu)
Kumanan R Wilson (Kumanan.Wilson@uhn.on.ca)

Version: 4 Date: 9 Jan 2002

Authors Response to Reviewer

Title: Teaching Evidence Based Complementary and Alternative Medicine (EBCAM); Changing behaviours in the face of reticence: A cross-over trial

Authors:
Edward J. Mills (emills@ccnm.edu)
Taras T. Hollyer (thollyer@ccnm.edu)
Ron Saranchuk (rsaranchuk@ccnm.edu)
Kumanan R. Wilson (Kumanan.Wilson@uhn.on.ca)

Version 3. Date: January 8, 2002

Reviewer: Dr. Andrew Vickers

To whom it may concern;

We again thank Dr. Vickers for taking the time to thoroughly review the submitted article and are grateful for the comments he has made. We accept the recommendations and have made the necessary changes. Please find below our responses to the individual comments.

1. The initial paragraphs have been rewritten to more follow Dr. Vickers suggested format. The paragraphs are now more concise and precise

2. Please note that Confidence Intervals are reported for the difference between groups for both scores and confidence in reading research in Table 4.

3. We agree that an ANCOVA would similarly display the results. We have retained the t-tests as they most simply display that the results are insignificant. We have now displayed the t-test results in a more conventional format and done away with the excess information, such as equality of variances details.

4. We have rewritten several paragraphs in the discussion section to be more concise. We have also removed the related quotes.
Yours sincerely

Edward J. Mills, DPH
Kumanan R. Wilson, MD, MSc
Taras Hollyer, MSc
Ron Saranchuk, MEd, PhD